Author's response to reviews

Title: Learning aptitude, spatial orientation and cognitive flexibility tested in a virtual labyrinth after virtual stress induction

Authors:

marcel delahaye (marcel.delahaye@upkbs.ch)
patrick lemoine (patrick.lemoine@upkbs.ch)
shanique cartwright (cartwrights76@gmail.com)
johannes beck (johannes.beck@upkbs.ch)
gunnar deuring (gunnar.deuring@upkbs.ch)
marlon pflueger (marlon.pflueger@upkbs.ch)
marc graf (marc.graf@upkbs.ch)
henning hachtel (henning.hachtel@upkbs.ch)

Version: 5
Date: 12 June 2015

Author's response to reviews:

Only an answer to the Editors was required (done by Email).